Author's response to reviews

Title: Epidemiological evidence of higher susceptibility to vCJD in the young

Authors:

Pierre-Yves Boelle (boelle@u444.jussieu.fr)
Jean-Yves Cesbron (jean-yves.cesbron@ujf-grenoble.fr)
Alain-Jacques Valleron (valleron@u444.jussieu.fr)

Version: 4 Date: 15 July 2004

Author's response to reviews:

Required formatting changes have been made.